Author’s response to reviews

Title: Nutritional Status of Children 7-36 Months old from Millet Consuming Communities of Masindi District, Western Uganda

Authors:

EVYLINE ISINGOMA (bisingoma@gmail.com)

Samuel Mbugua (smmbugwa@yahoo.com)

Edward Karuri (egkaruri@gmail.com)

Version: 4 Date: 24 Jan 2019

Author’s response to reviews:

- The datasets originally uploaded have been withdrawn and a statement, "The datasets used and/or analysed during the current study are available from the corresponding author on reasonable request.” added.

- Table 2 has now been referenced to in the main text on line 145.

- Abstract: Line 50 has been modified. The world 'slightly' has been added between showed and higher.

- Statements on additional files referenced to have been removed from line 143, 153, 169, 170 and 177.

- 'Study' population and 'study' children on lines 153, 183 and Table 1 have been changed to 'studied'

- The word 'only' has been removed from line 179 and the phrase 'of these' added between 28.5% and reported in the statement.

- The word respectively has been added on line 199 after 8%

- Line 226 the word children has been removed and replaced with 'them'.

- Line 257, the word persistent has been added between in and stunting.

- Line 266, the word proper has been added between of and childcare.

- Line 276 and Table 6, the word health care has been combined to read healthcare.

- Line 218, discriminatory measures changed to 'discriminatory'.
-Line 287 the word timely has been added between for and introduction.

-Line 400, all information on additional files has been removed.